Author's response to reviews

Title: Interventions aimed at reducing problems in adult patients discharged from hospital to home: a systematic meta-review

Authors:

Patriek Mistiaen (p.mistiaen@nivel.nl)
Anneke L. Francke (a.francke@nivel.nl)
Else Poot (e.poot@levv.nl)

Version: 5 Date: 2 April 2007

Author’s response to reviews:

Dear,

Thanks for the comments.
1. A sentence was added that 'all authors read and approved the final manuscript'
2. The table section is now started with a heading 'Tables'
3. The additional files section is now moved beneath the Tables
4. All underlining in headings had been removed and changed into bold text.

sincerely,

Patriek Mistiaen